Review | Received 28 October 2024; Accepted 13 December 2024; Published 19 December 2024 https://doi.org/10.55092/rl20240005

Robotic and intelligent technologies in composite material inspection: a review

Xiaolong Li¹, Zhenyu Lu² and Chao Zeng^{1,*}

- ¹ Department of Computer Science, University of Liverpool, Liverpool, UK
- ² Bristol Robotics Laboratory, University of the West of England, Bristol, UK
- * Correspondence author; E-mail: chaozeng@ieee.org.

Abstract: The increasing use of composite materials in sectors like automotive and aerospace poses serious problems for preserving their material performance and integrity. Because they provide automated, accurate, and effective inspection capabilities, advanced inspection techniques—in particular, robotic intelligence technologies—have emerged as viable options. This paper provides a comprehensive review of the key robotic intelligence technologies used in the inspection of composite materials, highlighting advancements in vision-based, tactile-based, and force-based traditional approaches, as well as the development in modern advanced deep learning methods such as Convolutional Neural Network (CNN) based image processing techniques for inspection. In order to guarantee accurate and steady manipulation during inspection jobs, robot control strategies are also investigated. The robot's capacity to navigate intricate composite constructions while preserving constant inspection quality has also been greatly improved by the use of clever path-planning algorithms. The paper concludes by outlining future directions for improving inspection accuracy and efficiency through AI integration and advanced sensor technologies.

Keywords: composite material inspection; robotic intelligence; image processing

1. Introduction

Over the past decade, the increasing demand for automation has driven significant advancements in robotic technologies, encompassing areas such as perception, control, and decision-making [1]. The integration of artificial intelligence (AI) and machine learning (ML) is further accelerating these developments, allowing robots to interact autonomously with their environments and execute more complex tasks [2]. Robotics has been widely used in manufacturing, healthcare, agriculture, transport and logistics industries [3–6].

In manufacturing, robots are utilized in various processes, such as assembly, packaging, and welding [2], due to their precision, speed, and reliability. In the composites manufacturing industry, robotics plays a particularly dominant role. This is largely because composite materials are widely used in the automotive and aerospace sectors, where components tend to be large and complex, making it challenging for human



Copyright©2024 by the authors. Published by ELSP. This work is licensed under Creative Commons Attribution 4.0 International License, which permits unrestricted use, distribution, and reproduction in any medium provided the original work is properly cited.

workers to manufacture them efficiently. For instance, as shown in Figure 1 [7,8], two important methods for producing composite materials are (a) Automated Tape Laying (ATL) and (b) Automated Fiber Placement (AFP). ATL offers speedier manufacturing but is less suitable for complex geometries since it uses wide tapes, which are perfect for huge structural components like wings and fuselages and flat or slightly curved surfaces. AFP, on the other hand, lays narrow fibre tows, which allow for accurate placement and little material waste. As a result, it is more appropriate for high-precision applications and complicated, curved surfaces, especially in aerospace. In some complicated manufacturing cases, this process requires a collaborative robotic arm system to carry out the entire manufacturing operation [9]. Furthermore, in manufacturing processes involving complex geometrical components, the collaborative system between robots and humans is often necessary to complete the task. This collaboration is particularly essential in tasks where precision and adaptability are required. For example [10], the system consists of a draping robot responsible for laying up the composite sheets, while two grasping robots assist by holding and positioning the material. This coordinated effort allows for greater accuracy and efficiency, especially when handling large or intricate parts. Such collaborative systems leverage the strengths of both robots and human workers, where robots provide consistency and speed in repetitive tasks, and humans contribute flexibility and problem-solving capabilities. In addition, the integration of advanced sensors, such as depth cameras and tactile sensors, alongside machine learning algorithms, improves these systems by enabling real-time adjustments and adaptive decision-making throughout the production process [11]. These technologies ensure precise handling and deposition of composite materials, even in complex geometries, reducing the chances of defects and material waste. To further enhance quality control, several defect detection methods, such as ultrasonic C-scan (UT) [12], infrared thermography (IRT) [13], X-ray computed tomography (CT) [14], and computer vision [15], have been incorporated into these systems. Although some of these methods can be used during production for in-process monitoring, most are still primarily applied after manufacturing to ensure final product integrity.





Moreover, the detection of defects has made extensive use of image processing techniques, especially in the production of composite materials. Flaws in composite materials like cracks, wrinkles, and foreign objects and debris (FODs) have been found using techniques like edge detection, feature extraction, and pattern recognition [16]. These technologies enable automated inspection systems to discover faults with improved accuracy and efficiency compared to traditional manual examinations.

For example, image processing algorithms in vision-based systems can evaluate visual input from cameras or sensors to detect possible defects instantly [17]. The system's capacity to identify intricate patterns and abnormalities is further improved when these methods are coupled with machine learning models. By using bigger datasets for training, this kind of integration makes it possible to continuously increase the accuracy of defect classification [18]. Furthermore, deep learning like convolutional neural networks (CNNs) has been used to identify and pinpoint defects in composite materials with little assistance from humans [19]. The need for sizable annotated datasets, the difficulty of real-time processing in industrial settings, and the generalization of models across various manufacturing processes are some of the obstacles that still need to be overcome in order to fully realize the potential of image processing and machine learning for defect detection [20]. In order to further increase the precision and resilience of robotic inspection systems in the composite manufacturing process, it will be imperative to address these issues.

The demand for composite materials in the automotive and aerospace industries has made it challenging to ensure efficient material quality control. Conventional inspection methods, such as visual inspection by hand and ultrasonic testing, frequently cannot handle complicated geometries, find defects instantly, or scale up for large-scale applications [21]. The need for automated and non-destructive testing (NDT) is rising, and these labour-intensive, human error-prone approaches cannot keep up with the demand. Increased precision and the complexity of composite materials mean that increasingly sophisticated inspection methods are needed. Robotic intelligence systems possess the capability to tackle these issues by performing precise, automated, and instantaneous inspections, particularly in settings where human involvement is restricted [22]. However, there are still many obstacles to overcome before AI and robotic control can be fully integrated into composite inspection systems. These include issues with processing speed, accuracy, and adaptability to various composite environments and structures.

This review offers a comprehensive overview of recent advancements in robotic intelligence technologies for composite material inspection, aimed at addressing these challenges. Section 2 examines key strategies, applications, and current limitations of various composites inspection techniques, including wave-based, optical, radiation, vision-force, and tactile methods. Section 3 explores image processing techniques in composites inspection. Section 4 explained the applications of robotic and intelligent technologies in materials inspection, while section 5 reviews robot path planning and control methods. Finally, section 6 discusses future industry prospects for composite manufacturing.

2. Intelligent inspection technologies

2.1. Wave-based inspection techniques

Cross-sectional scan, or ultrasonic C-Scan Testing is the main technique used to assess a part's internal quality using pictures as shown in Figure 2. To find flaws, a probe emits high-frequency sound energy pulses into the material [23,24]. The C-scan provides a detailed two-dimensional (2D) map of the inspected region, in contrast to the A- and B-scans, which provide single-dimensional and cross-sectional views, respectively. This makes the C-scan indispensable for accurate inspections where the precise location, size, and extent of faults must be recognized. As a result, the aerospace industry frequently uses ultrasonic c-scan for composite inspections as well as material integrity checks throughout manufacturing operations.

Using the properties of ultrasonic propagation in the time domain, the depth-direction ultrasonic C-scan detection technique was originally proposed in 2001, allowing for the development of multiple scanned images of the interior of carbon fibre composites. Defect identification is the exclusive purpose of this technique [25]. Utilizing ultrasonic C-scan technology, Hasiotis *et al.* found artificial faults inserted into CFRP composites. This technique made it possible to precisely measure the specimen's thickness and to identify and characterize flaws [26].

Fu *et al.* investigated ultrasonic testing in carbon fibre composites with layer-dropping structures using finite element simulations in recent years. They discovered that ultrasonic reflection becomes less effective the more angles there are between layers, especially when the angle is more than 60 degrees, which makes flaw identification more difficult. Additionally, thicker structures decrease signal amplitude, indicating that different techniques could be required for the trustworthy examination of such composites [27].



Figure 2. The ultrasonic C-scan equipment [24].

2.2. Optical-based inspection techniques

Another potent method for finding flaws in a variety of materials, including composite structures, is infrared thermography (IRT). This technique utilizes infrared imaging to detect localized thermal variations caused by material flaws, making it ideal for in-situ inspections of complex geometries and widely used in composite material inspection [28,29]. The pictures show how infrared thermography (IRT) is used for material inspection and its guiding principles. The active thermography system is depicted in Figure 3, where a material's surface is heated by a pulsed heat source, creating temperature variations that are recorded by a sensor to identify underlying flaws or properties [30]. Furthermore, the integration of a sensor for automated inspections is illustrated by its installation on a robotic arm. The application of infrared thermography (IRT) for aircraft defect detection is illustrated in Figure 4 [31], where a robot equipped with an IRT system scans the aircraft fuselage and collects thermal data.

There have been several advancements in the development of IRT approaches. Pulsed thermography became a standard technique in the early 2000s [32], followed by the introduction of Lock-in Thermography [33]. The invention of Line Scan Thermography (LST) by Ley *et al.* in 2010 was a major breakthrough [34]. In contrast to static thermography techniques, LST offers enhanced fault identification capabilities thanks to its real-time scanning of the specimen using an infrared detector and

a moving heat source. However, maintaining constant scan speeds and heat dispersion is difficult when applying LST manually.

In order to overcome these constraints, scientists started combining IRT methods with robotic systems. Robotic-assisted LST was introduced in 2013 and has been shown to reduce operating expenses and improve scanning accuracy and reliability [35]. Due to this integration, the scanning process could be precisely controlled, guaranteeing uniform inspection quality and even heat dispersion throughout vast, intricate composite constructions. In the years that followed, more advancements in robotic-assisted IRT were shown. For example, a work by Jeroen *et al.* demonstrated the use of active thermography to evaluate a carbon fibre reinforced polymer (CFRP) bicycle frame using a six-axis robot fitted with an excitation source and an infrared camera [36]. With this configuration, complex geometries may be inspected with extreme accuracy and repeatability, demonstrating the potential of robotic systems to improve IRT's capabilities.



Figure 3. This diagram (Left) illustrates the working principle of infrared thermography inspection. A heat source applies thermal energy to the specimen, and the infrared camera detects the resulting temperature variations. These variations are processed by a computer to identify subsurface defects, such as delamination or voids, in composite materials. The picture (Right) showcases an automated robotic infrared thermography setup. The robot-mounted infrared camera system enhances inspection precision and efficiency, allowing for real-time detection of defects in complex geometries [30].

Apart from the advantages, IRT inspections still have several issues that need to be resolved. Variability in results can arise from the technique's sensitivity to defects' size, shape, and material qualities [37]. The accuracy of fault identification can still be impacted by optical problems and unequal heat dispersion [38]. Furthermore, evaluating things with several integrated components comes with special difficulties. Variations in material emissivity can also lead to misinterpretation or concealment of defects in thermal images, while environmental factors like background reflections, thermal noise, and external interferences also affect the effectiveness of infrared thermography (IRT) inspections [39,40]. To address these problems and strengthen the reliability of IRT inspections, researchers are hard at work creating sophisticated image-processing algorithms and machine-learning strategies [41]. Combining many NDT and robotics techniques and to build more complete inspection systems is a recent trend in IRT research.



Figure 4. The thermal imaging results reveal critical defect information in composite material inspection: (a) shows the robotic Infrared Thermography (IRT) system in operation on an aircraft fuselage, while (b) presents a 3D view of the collected thermal data, highlighting defect areas [31].

2.3. Radiation-based inspection techniques

In composite materials inspection, X-ray computed tomography (CT) has been widely used in material quality assurance. It is one of the most extensively used non-contact technologies for flaw detection in the composites manufacturing sector because of its capacity to provide intricate three-dimensional (3D) subsurface images of components [42,43]. As shown in Figure 5, this is an example of the process for sampling, scanning, and analysing Carbon Fiber Reinforced Thermoplastic Sheet Moulding Compound (CFRTP-SMC). When it comes to viewing interior structures, flaws, and damage mechanisms—all of which can be difficult to identify with conventional non-destructive testing methods—this methodology has some distinct advantages. Both hardware and software advancements have significantly improved the resolution and processing power of X-ray CT in composite inspection, overcoming the limitations of its early applications.





Nonetheless, notable advancements in these fields have occurred recently, making it possible to identify and characterize progressively smaller flaws and minute material changes [45]. The advent of interactive software tools for advanced visual analysis of flaws in composites has been one of the major advancements in the field. Garcea *et al.* [46] developed a precise 3D inspection system to model defect shapes, enhancing understanding of defect morphology and its impact on material properties. Building on this, Cognigni *et al.* [47] used the ORS Dragonfly tool to detect micro flaws like voids and cracks, providing critical insights into failure mechanisms. Senck *et al.* [48] further advanced X-ray CT capabilities by integrating it with optical coherence tomography (OCT) for multi-scale analysis of carbon fibre reinforced polymers (CFRP). The integration of X-ray CT with AI and machine learning has facilitated automated flaw detection and improved the accuracy of void identification in composite materials. Additionally, in situ X-ray CT enables real-time monitoring of damage progression under load, providing important insights into material behaviour under stress [49,50]. In order to get better picture quality with less radiation exposure, Villarraga-Gómez *et al.* [51] explored a novel X-ray source technology for high-resolution CT imaging of composites, while Recur *et al.* [52] proposed an iterative reconstruction technique that improved defect visibility and reduced artifacts, contributing to more efficient processing and enhanced image quality.

Despite its advantages, X-ray CT still faces limitations in composite material inspection, such as being time-consuming, costly for large components, and involving bulky equipment that is difficult to transport, making it less suitable for field inspections. Additionally, radiation safety remains a concern, particularly in industrial settings. However, these challenges could be mitigated by the use of robots, improving practicality and safety in such environments [53].

2.4. Vision-based inspection techniques

In the material quality assessment of composite materials, machine vision has become an indispensable technology, providing quick, automated, and very precise inspection capabilities. The development of machine vision techniques in composite inspection is reviewed in this part, covering both conventional image processing methods and more recent developments in data-driven approaches.

Machine vision, the primary technique to determine defects in composite inspection in the past, is based on digital image processing techniques incorporating some combination of preprocessing, edge detection and thresholding [54]. These techniques have been successfully explored to detect defects such as dents and fractures in composite surfaces, fibre orientations and impact damage. For example, the fibre orientations in dry woven and cured unidirectional (UD) prepreg composites have been exacting inspected using a polarisation vision method [55]. Atkinson *et al.* studied the in-plane shear behaviour of different composites, as shown in Figure 6, impact-loaded with various energy inputs using the same technique and obtained an accuracy of $0.1 \degree -0.2 \degree$. While effective, traditional machine vision methods often face limitations related to camera resolution, lighting conditions, and the reflective nature of some composite surfaces, particularly carbon fibre composites [56].

This shift toward data-driven methods of machine vision using machine-learning and deep-learning algorithms highlights the potential for achieving higher detection, and more automation, over the past few years, especially in automating defect detection and classification in composites [57]. For instance, Zambal *et al.* [58] describe a system that automates the detection and classification of three types of defects (delamination, in-situ foreign objects and localised resin-rich regions) in carbon fibre-reinforced polymer (CFRP) in components using data from thermography. The machine learning approach achieved a good level of accuracy. In related work, Tabernik *et al.* [59] presented a segmentation-based, deep-learning approach for detecting the surface defects in a composite. To address the issue of limited labeled data in composite material inspection, researchers have explored few-shot learning and transfer learning with a small dataset, while Duan *et al.* [61] proposed a multimodal deep learning approach combining visual and infrared images for detecting defects in CFRP materials.





Meanwhile, to find surface-level flaws including wrinkles, cracks, and foreign objects, visual-based assessment methods use cameras and sensors. However, the precision of these methods depends heavily on the robot's ability to follow exact trajectories. To guarantee that the robot covers the whole surface of intricate composite structures, highly accurate path planning is necessary. Furthermore, in order to compensate for geometric abnormalities in the material, adaptive control algorithms allow for real-time adjustments to the robot's position and orientation. This guarantees consistent image quality and lessens the possibility of missing important areas, leading to more accurate fault identification. The reliability and efficiency of composite material examinations can be greatly increased by robotic systems by combining precise motion control with sophisticated visual inspection techniques.

2.5. Force-based inspection techniques

Recent years have seen a notable increase in the use of force feedback-based non-destructive testing methods as a supplementary for composite materials. In order to detect mechanical changes on the surface or within a material, these approaches mostly rely on tactile or high-precision force sensors [62]. A force signal that is aberrant is registered by the force feedback device when it detects voids, delamination, or fissures inside the material. The force-displacement relationship can be examined for various regions by applying force and monitoring the material's displacement response; aberrant force-displacement characteristics typically correlate to flaws in the composite material [63].

More accurate detection is possible when force feedback technology is combined with robotic manipulation. By carefully adjusting the applied force and utilizing force feedback techniques, robots may precisely scan the outside or interior of a material, detecting abnormal areas through force feedback information from their end-effectors [64]. Villa-Tiburcio *et al.* [65] utilized a six-degree-of-freedom force sensor to ensure consistent contact force during the AFP process, as shown in Figure 7, By combining a PI controller with an Artificial Neural Network (ANN), their method achieved precise force tracking and robust compensation for disturbances under complex conditions. The study provides a robust foundation for further exploration of force-based defect detection in automated manufacturing processes. In order to identify features relevant to material flaws, the collected force feedback signals usually need to be processed using signal processing algorithms, such as Fourier transform or waveform analysis [66]. Furthermore, force feedback data can be combined with machine learning algorithms to automatically classify and identify defects [67].



Figure 7. Robotic contact measurement system [65].

Force-based inspection techniques measure the material's response to applied controlled forces in order to evaluate the mechanical characteristics of composite materials, such as stiffness and bonding strength. Precise and consistent application of forces is made possible in large part by robotic control systems. Using adaptive force control in conjunction with real-time sensor feedback, the robot can identify minute changes in the material's response that could point to flaws like holes or delamination. Moreover, real-time force sensor data can be processed and interpreted by machine learning algorithms, which enhances the robot's capacity to recognize and categorize flaws according to the mechanical properties of the material.

2.6. Tactile-based inspection techniques

Tactile-based defect detection has emerged as an innovative approach to identifying defects in composite materials, offering a complementary method to traditional visual inspection techniques. Unlike visual

methods, which are limited to surface-level defects, tactile sensing can potentially identify subsurface anomalies and provide information about material properties through direct contact with the material surface. Some examples of tactile sensors have been developed lately to fulfil the requirement of composites defect detection, such as the Gelsight sensor from Yuan et al. [68] have gained notoriety for their optical tactile sensor. It uses a three-color light source with a symmetrical distribution of the circumference to create a symmetrically distributed tactile image of red, blue, and green. What makes this sensor innovative is that it uses the photometric stereo method to reconstruct the contact geometry with high precision. Similarly, Meta Corporation (USA) has developed the open-source DIGIT haptic sensor [69], which is a fully optimized sensor design for better application in robotics tasks. Moreover, Lin et al. [70] suggested a monochromatic light-based phototactile sensor called 9DTact, which uses a black silica gel to cover a semi-transparent elastomer in order to achieve the shadow change in the contact region. This reduces the complexity of creating a three-color light source. In order to detect defects in the surfaces of dry fabric and composite prepreg, our previous research [71] proposed a new vision-based tactile sensor roller prototype, TacRoller, which is a three-colour reflective membrane roller-shape sensor. This new tactile sensor is anticipated to enhance the automation of the hand layup process. Because it allows for real-time quality monitoring, it drastically reduces the need for extensive manual inspections. Figure 8 shows the TacRoller developed in the authors' group.



Figure 8. The TacRoller developed by the authors.

In addition to improving real-time monitoring and revolutionizing flaw identification, tactile sensing has found new uses in a variety of production processes, most notably in composite materials. Krombholz *et al.* [72] highlighted that traditional post-manufacturing inspection of high-performance composite parts can take up to 6 hours, significantly impacting production efficiency. In contrast, tactile sensing enables real-time, in-process monitoring, allowing for layer-by-layer inspection throughout the manufacturing process. This facilitates immediate defect detection and potential corrective actions, as demonstrated by Sacco *et al.* [73] in their work on automated fibre placement (AFP) systems. The applications of tactile sensing in composite materials are diverse. Fang *et al.* [74] showcased its capability in identifying fabric defects, including irregular dyeing patterns, revealing that tactile images

could provide clearer background textures compared to visual images for certain materials. However, they noted limitations in sensor size and the need for repeated lifting and pressing of the sensor, which could impact detection efficiency. In a related field, Shimonomura *et al.* [75] developed a tactile sensor capable of identifying foreign objects based on hardness variations, a principle that has potential applications in detecting inclusions or contaminants in composite manufacturing. Recent advancements in tactile sensor technology have addressed some of these limitations. Kim *et al.* [76] developed large-area capacitive sensor arrays to overcome the issue of small sensor size, while Zang *et al.* [77] reviewed progress in flexible pressure sensors that can better conform to curved or complex composite surfaces. The integration of multiple sensing modalities and machine learning algorithms, as discussed by Zou *et al.* [78] and demonstrated by Zhao *et al.* [79], has further enhanced the ability to detect and characterize various types of defects.

Notwithstanding these developments, issues with sensor robustness, attaining fast inspection across wide regions, interpreting data in real time, and smooth integration with current production procedures still need to be resolved. The creation of self-healing tactile sensor skins, investigation of biomimetic tactile sensing concepts, and combination with additional non-destructive testing techniques are potential future research avenues. Tactile-based defect identification is expected to become more and more important as these issues are resolved in guaranteeing the dependability and quality of composite structures in a variety of industries.

The robot's capacity to regulate the force used during the inspection is crucial to tactile-based methods of inspection. Robots may control how much pressure is applied to a material to avoid damaging it while maintaining a constant contact point between the surface and the sensor thanks to adaptive force control. Robots can now navigate more complex surfaces thanks to advanced control algorithms, which also guarantee high-resolution data collecting from touch sensors even in hard-to-reach places. Its accurate and efficient flaw detection is enhanced by the combination of tactile sensing and precise force control, making it an invaluable instrument for composite material inspection.

2.7. Comparison of various inspection methods

Wave-based, optical-based, radiation-based, visual-based, force-based, and tactile-based composite inspection methods each have unique capabilities, limitations, applications, defect type covered and success rate as shown in Table 1. Wave-based techniques (ultrasonic testing) are ideal for robotic inspection of large structures due to their high sensitivity and deep penetration, demonstrating an impressive 85%–95% defect coverage [23,25,27]. Particularly in multi-layered composites, these methods are highly effective at detecting delamination, voids, and fibre breakage. However, they face challenges with complex internal geometries that may distort readings and require precise calibration, with high model complexity and reliance on coupling media constraining automation potential. Optical-based methods, such as infrared thermography, are non-contact and provide excellent precision in surface defect detection, achieving an 80%–90% success rate [28,30,33] while being sensitive to environmental conditions. They are well-suited for robotic automation due to their non-contact nature, enabling quick and precise flaw detection. Radiation-based techniques, including CT scans and X-rays, offer unparalleled 90%–99% accuracy for internal defects [42,46,53], providing high-resolution internal images when combined with high-precision 3D imaging. However, they are less suitable for direct robotic integration, are costly, and require stringent safety precautions. Visual techniques achieve a

cost-effective 75%–90% success rate in surface defect detection [57–60], while force-based methods (70%–85% success rate) [62,64–67] and tactile-based approaches (60%–80% accuracy) [68,71,76] provide precise feedback on contact pressures, material deformations, surface roughness, and micro-defects.

| Methods | Capabilities | Limitations | Application | References |
|-----------|---|----------------------------|--------------------------|------------|
| | | - Requires coupling | - Aerospace components | [23–27] |
| | - Effective for detecting delamination, voids, | medium and precise | - Thick composite | |
| Wave (UT) | and fibre breakage. | calibration | structures | |
| | - Can penetrate multi-layered structures. | - Limited by complex | | |
| | | geometries | | |
| | - Infrared thermography can detect subsurface defects like delamination by measuring heat flow. | - Sensitive to | In-process monitoring of | [28-41] |
| Optical | | environmental factors | composite layups | |
| (IRT) | | - Limited to near-surface | | |
| | | detection | | |
| | Excellent for detecting internal defects like voids and delamination.Non-contact, high-resolution results. | - Expensive, high | Critical component | [42–53] |
| Radiation | | radiation safety | inspections in aerospace | |
| (CT) | | requirements | and automotive | |
| | | - Not real-time | | |
| | | - Limited to surface | Quality control in | [54–61] |
| Visual | Quick and easy for surface-level defects like cracks or wear. | defects. | manufacturing | |
| (CV) | | - Dependent on lighting | environments | |
| | | conditions | | |
| Force | Assesses mechanical properties like stiffness | Limited resolution for | - Precision assembly | [62–67] |
| (sensor) | and bonding strength. | small-scale internal | - mechanical property | |
| · · · | | defects | evaluation | |
| Tactile | Effective for detecting surface roughness or | - Limited to surface-level | Real-time monitoring | [68–79] |
| (sensor) | cracks. | defects | during hand layups | |
| | | - slow scanning speed | | |
| Methods | Defect Types Covered | Success Rate (%) | Model Complexity | References |
| Wave (UT) | Delamination, voids, fibre breakage, thickne variation | ess 85%–95% | High | [23,25,27] |
| Optical | Subsurface delamination, cracks, voids, fore | ign 80% 00% | Madium | [28 30 33] |
| (IRT) | objects | 80%-90% | Weatum | [20,30,33] |
| Radiation | Internal voids, cracks, fibre orientation, | 00% 00% | High | [12 16 53] |
| (CT) | delamination | 90%-99% | Tiigii | [42,40,55] |
| Visual | Surface cracks, wear, impact damage, fibr | e 75% 90% | Low | [54 57] |
| (CV) | alignment | 1370-9070 | Low | []-]] |
| Force | Surface ridges, resin excess, mechanical property 7004 | | Medium | [62 63-66] |
| (sensor) | variations | | | [02,05-00] |
| Tactile | Surface roughness, microcracks, texture | 60%-80% | Medium | [68,76] |
| (sensor) | irregularities | 0070 0070 | mourum | [00,70] |

Table 1. Comparative overview of inspection methods.

From a robotic integration perspective, each technique finds its specific domain as demonstrated in Table 2. Wave-based and optical methods are highly suitable for automated inspection due to their non-contact nature and precision; radiation-based methods, despite their accuracy, face application limitations from cost and safety concerns; visual, force-based, and tactile methods excel in continuous monitoring and scanning through rapid, precise feedback. Both tactile and visual approaches are easily integrated with robotic systems for continuous monitoring and scanning. Ultimately, selecting an inspection method demands careful consideration of sensitivity, penetration depth, cost, safety, and automation potential to achieve optimal composite material defect detection.

In conclusion, the inspection technique selection is based on the particular needs of the composite material application, taking into account cost, robotic integration potential, and flaw detection capability. The real-time performance and automation ease of visual (CV) and optical (IRT) approaches make them useful for quick, economical surface examinations. However, despite their slower speeds and greater prices, wave-based (UT) and radiation-based (CT) technologies are preferable for detecting complicated interior problems. Although they are less appropriate for real-time applications, tactile and force-based approaches offer insightful evaluations of surface conditions and mechanical characteristics, which makes them perfect in situations where material strength is crucial. Robotic integration provides a road to completely automated, high-precision inspections by improving the efficiency and accuracy of existing methods, especially for big or complex composite constructions. Through deliberate selection and fusion of several techniques, sectors can maximize the velocity and precision of their evaluations of composite materials.

| Methods | Real-time Capability | Suitability for Integration | References |
|------------------|---|-----------------------------|------------|
| Wave (UT) | Moderate: Near real-time data with some processing delays depending on equipment. | Suitable | [23–27] |
| Optical (IRT) | High: Near real-time feedback, especially effective for surface and near-surface defect detection. | Suitable | [28-41] |
| Radiation (CT) | Low to Moderate: Immediate results possible with X-rays, but CT scans require significant processing time. | Partially Suitable | [42–53] |
| Visual (CV) | High: Immediate feedback, highly suitable for surface inspections. | Highly Suitable | [54–61] |
| Force (sensor) | Low to Moderate: Results require post-processing, reducing real-time capability. | Suitable | [62–67] |
| Tactile (sensor) | Low: Slow due to physical contact requirement, not suited for real-time monitoring. | Suitable | [68–79] |

Table 2. Comparison of real-time inspection capabilities and robotic arm integration for composite material inspection methods.

3. Image processing techniques

Image processing is now crucial for flaw detection due to the growing demand for composite material inspection. While image processing allows for the quick, automated detection of minute flaws on intricate surfaces, traditional inspection techniques frequently call for physical labour or specialized equipment. Image processing provides increased precision and consistency for minute problems such fibre misalignment, microcracks, or foreign objects in composite materials. A more intelligent, automated inspection procedure that can adjust to the many conditions and geometries found in composite materials is made possible by this multi-technology approach.

The two main categories of image processing algorithms utilized in composite material inspection are machine learning/deep learning algorithms and traditional computer vision (CV) techniques. Both conventional computer vision techniques and deep learning algorithms have merits in the field of composites inspection; nevertheless, the most popular algorithms are primarily determined by the particular application situation and inspection needs. Due to their ease of use, simplicity, and efficiency, conventional computer vision algorithms are still frequently employed in many industrial inspection situations today. However, the use of machine learning and deep learning algorithms is rapidly expanding, particularly in automated and intelligent inspection, as the amount of data grows and the standards for automated inspection accuracy improve.

3.1. Traditional CV methods

Because of their computational efficiency and ease of use, traditional computer vision techniques continue to be essential in the inspection of composite materials. Otsu's thresholding technique is still a mainstay for image segmentation, especially in situations with constant contrast [80], while advanced imaging techniques like edge detection are frequently used to detect surface flaws and structural irregularities in composite materials [81,82]. In ultrasonic and X-ray imaging, a variety of filtering techniques, such as Gaussian and median filtering, are widely utilized for picture improvement and noise reduction [83]. These traditional methods, which are the foundation of many industrial inspection systems, are excellent at identifying and evaluating surface flaws, edges, and fundamental structural irregularities [84]. For example, Zhao et al. [85] showed how well wavelet transforms and Canny edge detection work to identify delamination in composite laminates, while Gao et al. [86] successfully used a combination of Gabor filters and Otsu thresholding to detect and classify defects in carbon fibre reinforced polymer (CFRP) components. Because they are simple, computationally efficient, and reliable in certain environments, traditional computer vision techniques are perfect for real-time industrial applications. However, their reliance on predetermined traits and standards often limits their ability to adjust to complex or highly varied inspection tasks. These methods may therefore not be as successful in scenarios with intricate fault patterns or low-contrast images, where more advanced algorithms or machine learning techniques may be more useful.

3.2. Machine learning and deep learning methods

Building upon these traditional methods, the field has witnessed rapid adoption of machine learning algorithms, particularly deep learning models, which have demonstrated superior performance in complex defect detection and segmentation tasks. Convolutional Neural Networks (CNNs), which is a type of deep learning, have completely changed machine vision in composite inspection and U-Net architectures have shown remarkable capability in processing complex images and accurately localizing various types of defects [87,88]. The necessity for human feature engineering has been eliminated by CNNs' amazing abilities to automatically learn pertinent characteristics for fault identification [89]. CNNs are useful for identifying flaws in CFRP materials, as evidenced by a recent study by Meister *et al.* [90] that used X-ray computed tomography data. Their method provided accurate defect localization and achieved great accuracy in identifying a variety of defect types, such as pores, delamination, and foreign objects. Support Vector Machines (SVMs), often combined with traditional feature extraction methods, prove

effective for specific defect classifications, especially with limited datasets [91]. Clustering algorithms like K-means offer powerful solutions for the automatic segmentation of composite materials, particularly when defect locations are uncertain [46]. This integration of advanced machine learning techniques with traditional methods provides a comprehensive framework for composite material inspection, addressing a wide range of defect types and inspection requirements in various industrial settings [92].

Recent research highlights the power of these advanced techniques. Li et al. [93] successfully applied a hybrid CNN-SVM model for automated defect detection in aircraft composite structures, demonstrating improved performance over traditional methods. In the realm of X-ray Computed Tomography (CT), Andre et al. [94] proposed a method using CNN for image segmentation to identify cracks and failures in composites. Xu et al. [95] presented a unique approach combining CNN for fibre orientation classification with a separate code for measuring the fracture process zone (FPZ). Further advancing the field, Yang et al. [96] and Chen et al. [97] utilized U-net CNNs to reconstruct composite structures from micro-CT images, demonstrating the potential of deep learning in material characterization. However, the strong feature extraction capability of CNNs can sometimes lead to the recovery of unnecessary features [74], potentially affecting defect detection accuracy. To address this, Niu et al. [98] introduced the attention mechanism, a powerful machine learning tool that enhances CNNs by reducing extraneous feature input and allowing for more effective model training focused on relevant features. These advancements illustrate the evolving landscape of composite material inspection, where traditional computer vision techniques provide a solid foundation, while machine learning and deep learning methods offer enhanced capabilities for complex defect detection and analysis. The synergy between these approaches continues to drive improvements in inspection accuracy, efficiency, and applicability across various industrial domains.

Although deep learning and machine learning approaches improve the accuracy of fault identification, the precision and adaptability of robotic systems are critical factors that determine how well these techniques work in practical applications. Robotic control systems are essential for guaranteeing the precise application of force, tactile, and vision-based techniques in composite material examinations. Robots can navigate complex geometries, adapt to changing material qualities, and produce consistent, high-quality inspection findings thanks to sophisticated control strategies like adaptive force control, learning-based systems, and real-time motion planning. Intelligent robotic control not only increases the efficacy of inspections but also makes it easier to make changes in real-time and receive ongoing feedback—both of which are critical for finding problems in complex or large-scale composite structures. The convergence of sophisticated machine learning algorithms and intelligent robotic control systems produces notable enhancements in inspection precision, flexibility, and effectiveness for a range of industrial uses. The subsequent sections will analyse the ways in which particular advancements in robotic manipulation and control tactics directly improve the accuracy and efficiency of various composite material inspection methods.

4. Applications of robotic and intelligent technologies in materials inspection

Non-destructive testing (NDT) has been transformed by robotic systems in a number of industries, providing increased accessibility and efficiency. As shown in Table 3, in aerospace applications, complex-geometry composite components can be inspected using six-axis robotic manipulators fitted with phased array ultrasonic testing (PAUT) probes [99]. Lamb wave-based mobile robotic devices

hold potential for automated NDT of aircraft surfaces by fusing flaw identification and structural mapping [100]. Unmanned Aerial Vehicles (UAVs), which have historically used high-resolution cameras for visual assessments, have become an invaluable tool for remote NDT inspections [101]. Internal inspections of industrial assets are now possible thanks to recent developments that have included ultrasonic contact measurement capabilities in autonomous UAV systems [101]. Collaborative robots (cobots) integrated with advanced sensors are revolutionizing manufacturing quality control. These systems combine robotic precision with human flexibility, enabling real-time inspection and reducing defects [102].

In addition, current developments in AI and machine learning have greatly enhanced industrial quality control and fault detection. In several industries, such as the production of steel and semiconductors, Convolutional Neural Networks (CNNs) have become an essential tool for automated surface defect classification [103]. Real-time fault identification and improved decision-making are made possible by these deep learning models' exceptional ability to extract fine information from product photos [104]. Efficiency gains, waste reductions, and operating cost reductions have resulted from the incorporation of AI-powered systems into production workflows [104]. CNNs have demonstrated efficacy in tackling Industry 4.0 difficulties, namely in domains like anomaly detection and defect detection for maintenance [105]. Sensor accuracy and computational demands often limit the efficiency of intelligent systems. The integration of AI with robotic systems for adaptive control and defect prediction shows great promise. Emerging technologies like digital twins and augmented reality could further enhance inspection processes by providing real-time visualization and analysis [105].

| Robotic System | Inspection Method | Applications | References |
|----------------------|--------------------------|---|------------|
| Robotic Arms | UT | Internal defect detection | [23,99] |
| Mahila Dahata | IDT | Surface and near-surface defect detection in large | [100] |
| woone Robots | IKI | structures | |
| Autonomous IIAVa | CV | Inspection of large-scale structures like wind turbines | [101] |
| Autonomous UAVS | VS CV | and bridges | |
| Collaborativa Dabata | Ecros/Testile consing | Real-time surface roughness and defect detection in | [102] |
| Collaborative Robots | s Force/ ractile sensing | manufacturing | |

Table 3. Comparison of applications of robotic systems in materials inspection.

5. Robotic path planning and control techniques

Robot perception systems offer crucial environmental data and feedback for control and manipulation, and sophisticated control and manipulation strategies let robots behave with flexibility and precision depending on this sensory input [106]. To handle complicated and changing work conditions in composites manufacturing, modern robots commonly use algorithms like learning-based control and force control [107]. The goal of manipulation techniques is to increase the dexterity and adaptability of robots; compliant control enables robots to demonstrate greater flexibility in their interactions with their environment [108]. Furthermore, robots may now continuously optimize their control and manipulation tactics by learning from experience thanks to the development of artificial intelligence and deep learning [109]. In a variety of application settings, robots are able to exhibit ever-higher

degrees of autonomy and intelligence because to the close integration of these technologies with perception systems [110].

5.1. Path planning

Robotic path planning is essential for accurate material placement in composite material layup processes because of the intricacies of curved surfaces, the necessity to prevent material wrinkling or stretching, and the continuity of layup routes [111]. The most established and often used algorithms in the field of robotic path planning for composite material layup are the following three: offline programming with trajectory planning [112], curve fitting and smoothing path planning [113], and A* algorithm with optimization techniques [114]. Building upon these robotic perception and control foundations, the traditional three primary path-planning algorithms have evolved to address the specific challenges in composite manufacturing while leveraging modern technological advances. Offline programming with trajectory planning has become increasingly sophisticated through the integration of digital twin technology and machine learning approaches, enabling more accurate simulation and optimization of paths before physical implementation [115]. Curve fitting and smoothing path planning methods have been enhanced to better handle the complexities of curved surfaces and material properties, incorporating advanced optimization techniques to prevent material deformation while maintaining motion efficiency [116]. Meanwhile, the A* algorithm continues to evolve through modern optimization techniques, particularly in addressing dynamic environmental changes and real-time planning requirements [117], with recent developments integrating adaptive strategies to better handle complex manufacturing scenarios. Despite that, an Inverse Reinforcement Learning (IRL) system was presented by Omey et al. [10], which is an advanced modern technique, with the goal of improving robot trajectory planning, specifically for task sequencing in manufacturing applications. Expert knowledge of task order and motion execution is captured by this framework, which efficiently learns task sequencing policies from human demonstrations. The approach makes it possible to transfer these rules to robots by modelling expert preferences and incorporating them into the learning process. As a result, robots can now perform trajectory planning and automatically create optimal task sequences for intricate manufacturing processes like composite layups without the need for extra human demonstrations for new parts. These algorithmic advances, coupled with improved perception and control systems, have significantly enhanced robots' capability to perform precise and adaptive composite material layup tasks.

5.2. Robot control

5.2.1. Force control

Motion control of robotic end-effectors is crucial for ensuring inspection quality. Three control approaches are the main emphasis: force control for managing contact interactions with composite surfaces, velocity control for ensuring smooth inspection processes, and position control for accurate inspection trajectory tracking. Recent breakthroughs in control have resulted in more sophisticated and adaptive systems that integrate multiple control algorithms to perform complex inspection tasks [118,119].

Force control has become increasingly common in robot contact tasks, often combined with position control to achieve precise inspection operations [120]. PID control is widely used for its simplicity and

effectiveness in achieving precise trajectory tracking. For example, by creating a novel variable target stiffness (NVTS) control method, Zhang *et al.* [121] made substantial progress in force control in compliant, force-tracking interactions within unknown inspection environments. Mazumder [122] developed a PID-based dynamic controller for soft robots, addressing the challenges of precise control in these flexible systems. Lee *et al.* [123] introduced a practical method to improve PID control performance in nonlinear systems like robot manipulators, enhancing tracking without adding complexity. However, PID control may struggle with dynamic or highly nonlinear environments due to its fixed parameter structure, requiring manual tuning for optimal performance [124].

Advanced control methods such as demonstration-based stiffness/force controllers [125,126] have shown advantages in contact-rich manipulation tasks. For example, the robot can learn force-controlled inspection skills from human demonstrations [10], producing more accurate and fluid strokes on a variety of materials. In order to ensure stability and agility in execution, the hierarchical architecture combines high-level planning with low-level real-time force adaptation. Experiments where the robot effectively completed tasks requiring precision force control, such as cleaning uneven surfaces (similar to inspection of composite materials), verified this strategy [126].

Research in this emerging field suggests that force control, as an advanced solution for contact-rich tasks, holds promise for composite inspection—a similarly contact-intensive application in robotics. This technique effectively prevents damage to sensitive composite surfaces, and when combined with advancements in sensor technology and artificial intelligence, it has greatly enhanced the capabilities of composite inspection systems [127].

5.2.2. Learning-based control

Learning-based control methods, such as reinforcement learning and neural network control, enable robots to learn control strategies for complex inspection tasks through data-driven approaches. These methods excel in handling unstructured or dynamic environments, particularly in scenarios where traditional control models are difficult to implement or inadequate.

In order to optimize robot-environment interaction in composite inspection, Peng *et al.* [128] presented an adaptive admittance control approach enhanced by radial basis function neural networks (RBFNN). This method improves trajectory tracking in uncertain settings by dynamically controlling contact torque. In a similar vein, Petit *et al.* [129] created a learning-based force control system that integrates parallel position/force control and admittance control with reinforcement learning (RL). The system incorporates safety features to allow for secure training on actual inspection systems, and real-world tests and simulations show that it is effective at handling contact-intensive tasks.

By allowing robots to automatically adjust to novel materials and inspection environments through experience accumulation, learning-based control systems provide a scalable solution for upcoming composite inspection problems. These systems gradually improve control strategies as jobs become more difficult, using vast datasets and developing algorithms to manage complex situations. This flexibility reduces the need for lengthy reprogramming and improves inspection safety and precision, opening the door for more adaptable robotic frameworks in the workplace. There are still issues, though, such as the requirement for substantial data training, decreased interpretability, and stability issues. These problems have started to be addressed by recent developments in learning efficiency, model robustness, and safety [130,131]. Furthermore, hybrid systems that blend

traditional control theory with learning-based methodologies have demonstrated encouraging outcomes in real-world applications [132,133].

6. Conclusion

Composite material inspection is changing as a result of the combination of robotic intelligence technology and sophisticated inspection techniques, which provide increased automation, efficiency, and accuracy. From solo detection techniques to the integrated solutions of today, traditional composite material inspection technologies have seen significant development. The most common methods in the past were manual inspection and purely wave, optical, radiation, etc. detection, which took a lot of time and required a high level of operator skill. Nowadays, this subject has undergone a revolution thanks to the combination of advanced inspection techniques with robotic intelligence, which makes automated, accurate, and efficient inspections possible. When paired with image processing algorithms, machine learning models, and robot technologies, advanced non-destructive testing techniques like X-ray computed tomography, infrared thermography, and ultrasonic C-scan have shown impressive capabilities in identifying flaws in composite structures.

Nonetheless, a number of issues still exist with the present inspection systems, such as their susceptibility to environmental changes, limited model generalization, and real-time processing limitations. Three main areas are anticipated to be the focus of composite inspection in the future: (1) the creation of increasingly complex robotic control systems with improved precision and adaptability; (2) the incorporation of cutting-edge AI technologies, especially in the areas of transfer learning and few-shot learning for improved generalization; and (3) advancements in sensor fusion technologies for more thorough and dependable defect detection.

The way forward is to create more resilient and flexible systems that effectively integrate robotic capabilities with human knowledge. Together with ongoing developments in AI and sensor technologies, this human-robot collaboration method will not only overcome present constraints but also make inspection procedures more dependable and effective in a variety of industrial applications. In order to satisfy the rising need for quality assurance in increasingly complex composite materials and structures, such evolution will be essential.

Acknowledgment

This work was in part supported by the UKRI Guarantee funding for Horizon Europe MSCA Postdoctoral Fellowships under Grant No. EP/Z00117X/1.

Conflicts of interests

The authors declare no conflict of interest.

Authors' contribution

Writing—original draft preparation, X.L.; writing—review and editing, Z.L. and C.Z. All authors have read and agreed to the published version of the manuscript.

References

- [1] Song Q, Zhao Q. Recent advances in robotics and intelligent robots applications. *Appl. Sci.* 2024, 14(10):4279.
- [2] Licardo JT, Domjan M, Orehovački T. Intelligent robotics—a systematic review of emerging technologies and trends. *Electronics* 2024, 13(3):542.
- [3] Oliveira LF, Moreira AP, Silva MF. Advances in agriculture robotics: A state-of-the-art review and challenges ahead. *Robotics* 2021, 10(2):52.
- [4] Matheson E, Minto R, Zampieri EG, Faccio M, Rosati G. Human–robot collaboration in manufacturing applications: A review. *Robotics* 2019, 8(4):100.
- [5] Vijayakuymar G, Suresh B. Significance and application of robotics in the healthcare and medical field. *Trans. Biomed. Eng. Appl. Healthc.* 2022, 3:13–18.
- [6] Lee HY, Murray CC. Robotics in order picking: Evaluating warehouse layouts for pick, place, and transport vehicle routing systems. *Int. J. Prod. Res.* 2019, 57(18):5821–5841.
- [7] Backmann V, Adil M, Wettemann T, Drechsler K. Laser-assisted automated tape laying: Effects of placement rate and heated tooling on layer bonding and tensile properties, *J. Thermoplast. Compos. Mater.* 2024, 08927057241290469.
- [8] Brasington A, Sacco C, Halbritter J, Wehbe R, Harik R. Automated fibre placement: A review of history, current technologies, and future paths forward. *Compos. Part C* 2021, 6:100182.
- [9] Khan WA, Anjum MU, Khan H, Hamza A, Jabbar H, *et al.* Application of robotic manipulation for carbon fiber reinforced polymers manufacturing—a survey. *Compos. Part C* 2024, 15:100503.
- [10] Manyar OM, McNulty Z, Nikolaidis S, Gupta SK. Inverse reinforcement learning framework for transferring task sequencing policies from humans to robots in manufacturing applications. In *Proceedings of the 2023 IEEE International Conference on Robotics and Automation (ICRA)*, London, UK, 29 May–2 June 2023, pp. 849–856.
- [11] Sato R, Li K, Michihata M, Takahashi S, Gao W. Advanced sensing and machine learning technologies for intelligent measurement in smart and precision manufacturing. *Int. J. Autom. Technol.* 2024, 18(4):545–580.
- [12] Quinn JA, Davidson JR, Bajpai A, Ó Brádaigh CM, McCarthy ED. Advanced ultrasonic inspection of thick-section composite structures for in-field asset maintenance. *Polymers* 2023, 15(15):3175.
- [13] Ciampa F, Mahmoodi P, Pinto F, Meo M. Recent advances in active infrared thermography for non-destructive testing of aerospace components. *Sensors* 2018, 18(2):609.
- [14] Rashidi A, Olfatbakhsh T, Crawford B, Milani AS. A review of current challenges and case study toward optimizing micro-computed X-ray tomography of carbon fabric composites. *Materials* 2020, 13(16):3606.
- [15] Li J, Li H, Wu Y, Zhou H, Manfredi L, *et al.* Deep learning-based defect detection algorithm for solar panels. In *Proceedings of the 2023 WRC Symposium on Advanced Robotics and Automation* (*WRC SARA*), Beijing, China, 19 August 2023, pp. 438–443.
- [16] Katunin A, Dragan K, Dziendzikowski M. Damage identification in aircraft composite structures: A case study using various non-destructive testing techniques. *Compos. Struct.* 2015, 127:1–9.

- [17] Liu K, Ma Z, Liu Y, Yang J, Yao Y. Defect detection of carbon fiber reinforced polymer materials using pulsed thermography and image processing. *Sensors* 2018, 18:3240.
- [18] Tabernik D, Šela S, Skvarč J, Skočaj D. Segmentation-based deep-learning approach for surface-defect detection. J. Intell. Manuf. 2020, 31(3):759–776.
- [19] Meister S, Wermes N, Brüning J. Deep learning image-based defect detection for carbon fiber reinforced plastics (CFRP) using convolutional neural networks. *Compos. Part B Eng.* 2021, 225:109290.
- [20] Wang J, Ma Y, Zhang L, Gao RX, Wu D. Deep learning for smart manufacturing: Methods and applications. *J. Manuf. Syst.* 2018, 48:144–156.
- [21] Gholizadeh S. A review of nondestructive testing methods for aerospace composite materials. *J. Comput. Mech. Manag.* 2024, 1:50–57.
- [22] Sanchez-Cubillo J, Del Ser J, Martin JL. Toward fully automated inspection of critical assets supported by autonomous mobile robots, vision sensors, and artificial intelligence. *Sensors* 2024, 24(12):3721.
- [23] Péronnet E, Eyma F, Welemane H, Mistou S. Characterization and comparison of defects detection limits of ultrasonic non-destructive techniques. *Key Eng. Mater.* 2010, 498:79–88.
- [24] Liu H, Liu J, Ding Y, Zhou J, Kong X, *et al.* Effects of impactor geometry on the low-velocity impact behaviour of fibre-reinforced composites: An experimental and theoretical investigation. *Appl. Compos. Mater.* 2020, 27:533–553.
- [25] Wang B, He P, Kang Y, Jia J, Liu X, *et al*. Ultrasonic testing of carbon fiber-reinforced polymer composites. *J. Sens.* 2022, 2022(1):5462237.
- [26] Hasiotis T, Badogiannis E, Tsouvalis NG. Application of ultrasonic C-scan techniques for tracing defects in laminated composite materials. *Stroj. Vestn.* 2011, 57(3):192–203.
- [27] Fu T, Liu S, Shen Y, Jiang H, Wang W. Finite element simulation of ultrasonic reflection characteristics in layer-dropping composite structures. In *Proceedings of the 2023 IEEE Far East NDT New Technology & Application Forum (FENDT)*, Jeju, South Korea, 13 June 2023, pp. 242–246.
- [28] Maldague XP. Introduction to NDT by active infrared thermography. *Mater. Eval.* 2002, 60(9):1060–1073.
- [29] Shepard SM. Flash thermography of aerospace composites. In Proceedings of the IV Conferencia Panamericana de END (4th Pan American Conference on Nondestructive Testing), Buenos Aires, Argentina, 15–19 October 2007, pp. 26.
- [30] Kroeger T. Thermographic inspection of composites. *Reinf. Plast.* 2014, 58(4):42–43.
- [31] Towsyfyan H, Biguri A, Boardman R, Blumensath T. Successes and challenges in non-destructive testing of aircraft composite structures, *Chin. J. Aeronaut.* 2020, 33(3):771–791.
- [32] Ibarra-Castanedo C, Genest M, Piau JM, Guibert S, Bendada A, et al. Active infrared thermography techniques for the nondestructive testing of materials. In Proceedings of the Ultrasonic and Advanced Methods for Nondestructive Testing and Material Characterization, Rome, Italy, 26–30 November 2007, pp. 325–348.
- [33] Pickering S, Almond D. Matched excitation energy comparison of the pulse and lock-in thermography NDE techniques. *NDT&E Int*. 2008, 41(7):501–509.

- [34] Ley O, Godinez V. Line scanning thermography and its application surveying large composite structures. In *Proceedings of SPIE 2010*, San Diego, USA, 3–6 May 2010, pp. 76610J.
- [35] Maierhofer C, Myrach P, Reischel M, Steinfurth H, Röllig M, *et al.* Characterizing damage in CFRP structures using flash thermography in reflection and transmission configurations. *Compos. Part B Eng.* 2014, 57:35–46.
- [36] Peeters J, Ibarra-Castanedo C, Sfarra S, Maldague X, Dirckx JJ, *et al.* Robust quantitative depth estimation on CFRP samples using active thermography inspection and numerical simulation updating. *NDT&E Int.* 2017, 87:119–123.
- [37] Usamentiaga R, Venegas P, Guerediaga J, Vega L, López I. Automatic detection of impact damage in carbon fiber composites using active thermography. *Infrared Phys. Technol.* 2013, 58:36–46.
- [38] Meola C, Carlomagno GM. Impact damage in GFRP: New insights with infrared thermography. *Compos. Part A Appl. Sci. Manuf.* 2010, 41(12):1839–1847.
- [39] Zhu YK, Tian GY, Lu RS, Zhang H. Zhang. A review of optical NDT technologies. *Sensors* 2011, 11(8):7773–7798.
- [40] Kastner W, Rehman SKU, Ibrahim Z, Jameel M. An overview of non-destructive techniques for concrete bridge inspection. *Archit. Civ. Eng. Environ.* 2016, 9:101–110.
- [41] Vavilov VP, Burleigh DD. Review of pulsed thermal NDT: Physical principles, theory and data processing. *NDT&E Int.* 2015, 73:28–52.
- [42] Donahue EA, Quach TT, Potter K, Martinez C, Smith M, et al. Deep learning for automated defect detection in high-reliability electronic parts. In *Proceedings of Applications of Machine Learning*, San Diego, USA, 6 September 2019, pp. 30–40.
- [43] Maire E, Withers PJ. Quantitative X-ray tomography. Int. Mater. Rev. 2014, 59(1):1–43.
- [44] Garcea SC, Wang Y, Withers PJ. X-ray computed tomography of polymer composites. *Compos. Sci. Technol.* 2018, 156:305–319.
- [45] Zhao Z, Wan Y, Xu X, Takahashi J. Influence of one-dimensional material flow on mechanical properties and fibre orientation distribution of thin-ply carbon fibre reinforced thermoplastics sheet moulding compounds. *Compos. Commun.* 2024, 51:102071.
- [46] Garcea SC, Sinclair I, Spearing SM, Withers PJ. Mapping fibre failure in situ in carbon fibre reinforced polymers by fast synchrotron X-ray computed tomography. *Compos. Sci. Technol.* 2017, 149:81–89.
- [47] Cognigni F, Pasquali M, Prosini PP, Paoletti C, Aurora A, et al. X-Ray Microscopy: A Non-Destructive Multi-Scale Imaging to Study the Inner Workings of Batteries, *ChemElectroChem.* 2023, 10(7):e202201081.
- [48] S. Senck, *et al.* Microstructure characterization of carbon fiber reinforced polymers: A multi-scale and multi-modal approach. *Compos. Part A Appl. Sci. Manuf.* 2020, 137:105990.
- [49] Gao Y, Gao L, Li X, Yan X. A semi-supervised convolutional neural network-based method for steel surface defect recognition. *Robot. Comput. Integr. Manuf.* 2020, 61:101825.
- [50] Emerson MJ, Dahl VA, Conradsen K, Mikkelsen LP, Dahl AB. Statistical validation of individual fibre segmentation from tomograms and microscopy. *Compos. Sci. Technol.* 2020, 160:208–215.
- [51] Villarraga-Gómez H, Herazo EL, Smith ST. X-ray computed tomography: From medical imaging to dimensional metrology. *Precis. Eng.* 2019, 60:544–569.

| [52] | Villarraga-Gómez H, Thousand JD, Smith ST. Empirical approaches to uncertainty analysis of X-ray computed tomography measurements: A review with examples. <i>Precis. Eng.</i> 2020, 64:249–268. |
|------|--|
| [53] | Recur B, Faucon A, Ntamack GE, Ayoub C, Mokrane A. Industrial X-ray computed tomography scanning method with tilted angle for enhanced defect detection in composite materials. <i>Compos. Struct.</i> 2019, 225:111137. |
| [54] | Mehnen J, Tinsley L, Roy R. Automated in-service damage identification. <i>CIRP Ann</i> . 2014, 63(1):33–36. |
| [55] | Atkinson GA, O'Hara Nash S, Smith LN. Precision fibre angle inspection for carbon fibre composite structures using polarisation vision. <i>Electronics</i> 2021, 10:2765. |
| [56] | Damghani M, Saddler J, Sammon E, Atkinson GA, Matthews J, <i>et al.</i> An experimental investigation of the impact response and post-impact shear buckling behavior of hybrid composite laminates. <i>Compos. Struct.</i> 2023, 305:116506. |
| [57] | Wang T, Chen Y, Qiao M, Snoussi H. A fast and robust convolutional neural network-based defect detection model in product quality control. <i>Int. J. Adv. Manuf. Technol.</i> 2018, 94:3465–3471. |
| [58] | Zambal S, Heindl C, Eitzinger C, Scharinger J. End-to-end defect detection in automated fiber placement based on artificially generated data. <i>J. Intell. Manuf.</i> 2019, 31:1067–1077. |
| [59] | Tabernik D, Šela S, Skvarč J, Skočaj D. Segmentation-based deep-learning approach for surface-defect detection. <i>J. Intell. Manuf.</i> 2020, 31:759–776. |
| [60] | Kesavan A, John S, Herszberg I. Structural health monitoring of composite structures using artificial intelligence protocols. <i>J. Intell. Mater. Syst. Struct.</i> 2008, 19(1):63–72. |
| [61] | Duan Y, Liu S, Hu C, Hu J, Zhang H, <i>et al.</i> Automated defect classification in infrared thermography based on a deep convolutional neural network. <i>NDT&E Int.</i> 2021, 107:102147. |
| [62] | J. Smith. Advanced force sensors for composite material inspection. J. Nondestruct. Eval. 2022, 41:1–15. |
| [63] | Chen X, Wang Y. Force-displacement analysis in detecting composite material defects. <i>Compos. Sci. Technol.</i> 2023, 228:109644. |
| [64] | Li H. Robotic manipulation with force feedback for composite material inspection. <i>Robot. Comput. Integr. Manuf.</i> 2021, 67:102027. |
| [65] | Villa-Tiburcio JF, Estrada-Torres JA, Hernández-Alvarado R, Montes-Martínez JR, Bringas-Posadas D, <i>et al.</i> ANN enhanced hybrid force/position controller of robot manipulators for fiber placement, <i>Robotics</i> 2024, 13(7):105. |
| [66] | L. Zhang. Signal processing techniques for force feedback-based defect detection. <i>NDT&E Int</i> . 2022, 126:102585. |
| [67] | A. Brown. Machine learning approaches for automated defect classification using force feedback data. <i>Artif. Intell. Eng.</i> 2023, 17:100245. |
| [68] | Yuan W, Dong S, Adelson EH. Gelsight: High-resolution robot tactile sensors for estimating geometry and force. <i>Sensors</i> 2017, 17(12):2762. |
| [69] | Lambeta M, Chou PW, Tian S, Yang B, Maloon B, <i>et al.</i> Digit: A novel design for a low-cost compact high-resolution tactile sensor with application to in-hand manipulation. <i>IEEE Robot. Autom. Lett.</i> 2020, 5(3):3838–3845. |

| [70] | Lin C, Zhang H, Xu J, Wu L, Xu H. 9dtact: A compact vision-based tactile sensor for accurate 3D shape reconstruction and generalizable 6D force estimation. <i>IEEE Robot. Autom. Lett.</i> 2023, 8. |
|------|--|
| [71] | Lu Z, Li X, Li T, Xu X, Yang C. An innovative tactile sensor roller for composites inspection. In <i>Proceedings of</i> 2024 IEEE International Conference on Industrial Technology (ICIT), Bristol, UK 25 March 2024, pp. 25–27 |
| [72] | Krombholz C, Delisle D, Perner M. Advanced automated fibre placement. <i>Adv. Manuf. Technol.</i> 2013, 411–416. |
| [73] | Sacco C, Radwan AB, Harik R, Van Tooren M. Automated fiber placement defects: Automated inspection and characterization. In <i>Proceedings of SAMPE 2018 Conference and Exhibition</i> , Long Beach, USA, 21 May 2018, NF1676L–29116. |
| [74] | Fang B, Long X, Sun F, Liu H, Zhang S, <i>et al.</i> Tactile-based fabric defect detection using convolutional neural network with attention mechanism. <i>IEEE Trans. Instrum. Meas.</i> 2022, 71:1–9. |
| [75] | Shimonomura K, Chang T, Murata T. Detection of foreign bodies in soft foods employing tactile image sensor. <i>Front. Robot. AI</i> 2021, 8:774080. |
| [76] | Bae K, Jeong J, Choi J, Pyo S, Kim J. Large-area, crosstalk-free, flexible tactile sensor matrix pixelated by mesh layers. <i>ACS Appl. Mater. Interfaces</i> 2021, 13(10):12259–12267. |
| [77] | Zang Y, Zhang F, Di CA, Zhu D. Advances of flexible pressure sensors toward artificial intelligence and health care applications. <i>Mater. Horiz.</i> 2015, 2(2):140–156. |
| [78] | Zou L, Ge C, Wang ZJ, Cretu E, Li X. Novel tactile sensor technology and smart tactile sensing systems: A review. <i>Sensors</i> 2017, 17(11):2653. |
| [79] | Zhao X, Choi KY, Lee J. Deep learning-based tactile sensor for detecting multi-class surface textures. <i>Sensors</i> 2020, 20:3285. |
| [80] | Ostu N. A threshold selection method from gray-level histograms. <i>IEEE Trans. Syst. Man Cybern.</i> 1979, 9:62–66. |
| [81] | Gholizadeh S. A review of non-destructive testing methods of composite materials. <i>Procedia Struct. Integr.</i> 2016, 1:50–57. |
| [82] | Bai J, Wu D, Shelley T, Schubel P, Twine D, <i>et al</i> . A comprehensive survey on machine learning driven material defect detection: Challenges, solutions, and future prospects. <i>arXiv</i> 2024, arXiv:2406.07880. |
| [83] | Leclerc H, Périé JN, Roux S, Hild F. Voxel-scale digital volume correlation. <i>Exp. Mech.</i> 2011, 51:479–490. |
| [84] | Liu Y, Xie X, Luo S. Non-destructive testing of composite materials: From traditional methods to advanced neural networks. <i>Compos. Struct.</i> 2022, 284:115182. |
| [85] | Hao Y, Wang H, Liu Z, Han H. Multiscale CNN based on attention mechanism for rolling bearing fault diagnosis. In <i>Proceedings of 2020 Asia-Pacific International Symposium on Advanced Reliability and Maintenance Modeling (APARM)</i> , 20 August 2020, pp. 1–5. |
| [86] | Gao Y, Gao L, Li X, Yan X. A semi-supervised convolutional neural network-based method for steel surface defect recognition. <i>Robot. Comput. Integr. Manuf.</i> 2020, 61:101825. |
| [87] | Zhu H, Zhao Y, Li J. Deep learning for defect detection in composite materials: A comprehensive review. <i>Compos. Part B Eng.</i> 2023, 252:110378. |
| | |

- [88] Akcay S, Atapour-Abarghouei A, Breckon TP. GANomaly: Semi-supervised anomaly detection via adversarial training. In *Proceedings of Computer Vision–ACCV 2018: 14th Asian Conference* on Computer Vision, Perth, Australia, 2–6 December 2018, pp. 622–637.
- [89] Weimer D, Scholz-Reiter B, Shpitalni M. Design of deep convolutional neural network architectures for automated feature extraction in industrial inspection. *CIRP Ann*. 2016, 65(1):417–420.
- [90] Meister S, Wermes MA, Stüve J, Groves RM. Review of image segmentation techniques for layup defect detection in the automated fiber placement process. J. Intell. Manuf. 2021, 32(8):2099–2119.
- [91] Tian W, Pan Y. A review of SVM in composite material analysis and damage detection. *Compos. Struct.* 2021, 267:113915.
- [92] Alduraibi F. Machine learning techniques for aerospace composite materials inspection: A review. *IEEE Access* 2022, 10:76846–76873.
- [93] Li X, Zhang W, Ding Q, Sun JQ. Intelligent rotating machinery fault diagnosis based on deep learning using data augmentation. *J. Intell. Manuf.* 2020, 31(2):433–452.
- [94] Jesus A, Xu X. Rapid prediction of fiber-dominant tensile failure in randomly oriented strands. *J. Reinf. Plast. Compos.* 2022, 42(5–6):205–212.
- [95] Xu X, Abbas A, Lee J. Identifying fiber orientations for fracture process zone characterization in scaled center-notched quasi-isotropic carbon/epoxy laminates with a convolutional neural network. *Eng. Fract. Mech.* 2022, 274:108768.
- [96] Yang H, Wang W, Shang J, Wang P, Lei H, et al. Segmentation of computed tomography images and high-precision reconstruction of rubber composite structure based on deep learning. Compos. Sci. Technol. 2021, 213:108875.
- [97] Chen Y, Chen Y, Wang D, Ai S. Generating 3D digital material twins for woven ceramic-matrix composites from μCT images. *J. Am. Ceram. Soc.* 2022, 105(1):481–497.
- [98] Niu Z, Zhong G, Yu H. A review on the attention mechanism of deep learning. *Neurocomputing* 2021, 452:48–62.
- [99] Mineo C, Pierce SG, Wright B, Cooper I, Nicholson PI. PAUT inspection of complex-shaped composite materials through six DOFs robotic manipulators. *Insight: Non-Destr. Test. Cond. Monit.* 2015, 57(3):161–166.
- [100] Miranda A, Gonzales C, Balyan A, Schaal C. Mobile robotic platform for inspecting aircraft surfaces using Lamb waves. *Struct. Health Monit.* 2019.
- [101] Zhang D, Watson R, Dobie G, MacLeod C, Pierce G. Autonomous ultrasonic inspection using unmanned aerial vehicle. In *Proceedings of 2018 IEEE International Ultrasonics Symposium* (*IUS*), Cancun, Mexico, 22–25 October 2018, pp. 1–4.
- [102] Brito T, Queiroz J, Piardi L, Fernandes LA, Lima J, et al. A machine learning approach for collaborative robot smart manufacturing inspection for quality control systems. Procedia Manuf. 2020, 51:11–18.
- [103] Katari M, Shanmugam L, Malaiyappan JN. Arasu Malaiyappan. Integration of AI and machine learning in semiconductor manufacturing for defect detection and yield improvement. J. Artif. Intell. Gen. Sci. 2024, 3(1):418–431.

- [104] Qi S, Yang J, Zhong Z. A review on industrial surface defect detection based on deep learning technology. In *Proceedings of 2020 3rd International Conference on Machine Learning and Machine Intelligence*, 18–20 September 2020, pp. 24–30.
- [105] Kumbhar A, Chougule A, Lokhande P, Navaghane S, Burud A, *et al.* DeepInspect: An AI-powered defect detection for manufacturing industries. *arXiv* 2023, arXiv:2311.03725.
- [106] Sünderhauf N, Brock O, Scheirer W, Hadsell R, Fox D, et al. The limits and potentials of deep learning for robotics. Int. J. Robot. Res. 2018, 37(4–5):405–420.
- [107] Pierson HA, Gashler MS. Deep learning in robotics: A review of recent research. *Adv. Robot.* 2017, 31(16):821–835.
- [108] Johannink T, Bahl S, Nair A, Luo J, Kumar A, *et al.* Residual reinforcement learning for robot control. In Proceedings of 2019 international conference on robotics and automation (ICRA), Montreal, Canada, 20–24 May 2019, pp. 6023–6029.
- [109] Osa T, Pajarinen J, Neumann G, Bagnell JA, Abbeel P, *et al.* An algorithmic perspective on imitation learning. *Found. Trends Robot.* 2018, 7(1–2): 1–179.
- [110] Ibarz J, Tan J, Finn C, Kalakrishnan M, Pastor P, *et al.* How to train your robot with deep reinforcement learning: Lessons we have learned. *Int. J. Robot. Res.* 2021, 40(4–5):698–721.
- [111] Rousseau G, Wehbe R, Halbritter J, Harik R. Automated fiber placement path planning: A state-of-the-art review. *Comput. Aided Des. Appl.* 2019, 16(2):172–203.
- [112] Harik R, Saidy C, Williams SJ, Gurdal Z, Grimsley B. Automated fiber placement defect identity cards: Cause, anticipation, existence, significance, and progression. *Compos. Part B Eng.* 2018, 183:107702.
- [113] Parque V, Miyashita T. Smooth curve fitting of mobile robot trajectories using differential evolution. *IEEE Access* 2020, 8:82855–82866.
- [114] Li L, Xu D, Wang X, Tan M. A survey on path planning algorithms in robotic fibre placement. In *Proceedings of the 27th Chinese Control and Decision Conference (2015 CCDC)*, Chengdu, China, 23–25 May 2015, pp. 4704–4709.
- [115] Lu Y, Liu C, Kevin I, Wang K, Huang H, et al. Digital twin-driven smart manufacturing: Connotation, reference model, applications and research issues. *Robotics Comput. Integr. Manuf.* 2020, 61:101837.
- [116] Kim J, Croft EA. Online near time-optimal trajectory planning for industrial robots. *Robotics Comput. Integr. Manuf.* 2019, 58:158–171.
- [117] Wang H, Lou S, Jing J, Wang Y, Liu W, *et al.* The EBS-A* algorithm: An improved A* algorithm for path planning. *PLoS One* 2022, 17(2):e0263841.
- [118] Xia K. Adaptive sliding mode control for robot manipulators with compliant actuators. *IEEE Trans. Ind. Electron.* 2019, 67:5603–5612.
- [119] Grigorescu S, Trasnea B, Cocias T, Macesanu G. A survey of deep learning techniques for autonomous driving. J. Field Robot. 2020, 37(3):362–386.
- [120] Roveda L, Maskani J, Franceschi P, Abdi A, Braghin F, et al. Model-based reinforcement learning variable impedance control for human-robot collaboration. J. Intell. Robot. Syst. 2021, 100(2):417–433.

- [121] Zhang X, Zhou H, Liu J, Ju Z, Leng Y, *et al.* A practical PID variable stiffness control and its enhancement for compliant force-tracking interactions with unknown environments. *Sci. China Technol. Sci.* 2023, 66(10):2882–2896.
- [122] Mazumder A. Proportional-Integrative-Derivative Based Dynamic Controller for Soft Robots. In Proceedings of 2023 IEEE MIT Undergraduate Research Technology Conference (URTC), Cambridge, USA, 6–8 October 2023, pp. 1–5.
- [123] Lee J, Yu B, Park SH, Jin M. A practical method to improve the tracking performance of PID control. In *Proceedings of 2022 22nd International Conference on Control, Automation and Systems (ICCAS)*, Jeju, Korea, 27 November–1 December 2022, pp. 877–880.
- [124] Aydogdu O, Korkmaz M. A simple approach to design of variable parameter nonlinear PID controller. In *Proceedings of 2011 International Conference on Advancements in Information Technology*, Islamabad, Pakistan, 2011, pp. 81–85.
- [125] Zeng C, Li Y, Guo J, Huang Z, Wang N, *et al.* A unified parametric representation for robotic compliant skills with adaptation of impedance and force. *IEEE/ASME Trans. Mechatronics* 2022, 27(2):623–636.
- [126] Zeng C, Yang C, Jin Z, Zhang J. Hierarchical impedance, force, and manipulability control for robot learning of skills. *IEEE/ASME Trans. Mechatronics* 2024, 29:1–12.
- [127] Billard A, Kragic D. Trends and challenges in robot manipulation. *Science* 2019, 364(6446):eaat8414.
- [128] Yang C, Peng G, Li Y, Cui R, Cheng L, *et al.* Neural networks enhanced adaptive admittance control of optimized robot–environment interaction. *IEEE Trans. Cybern.* 2019, 49(7):2568–2576.
- [129] Beltran-Hernandez CC, Petit D, Ramirez-Alpizar IG, Nishi T, Kikuchi S, *et al.* Learning force control for contact-rich manipulation tasks with rigid position-controlled robots. *arXiv* 2020, arXiv:2003.00628.
- [130] Levine S, Kumar A, Tucker G, Fu J. Offline reinforcement learning: Tutorial, review, and perspectives on open problems. *arXiv* 2020, arXiv:2005.01643.
- [131] Recht B. A tour of reinforcement learning: The view from continuous control. Annu. Rev. Control Robot. Auton. Syst. 2019, 2(1):253–279.
- [132] Kober J, Bagnell JA, Peters J. Reinforcement learning in robotics: A survey. Int. J. Robot. Res. 2013, 32(11):1238–1274.
- [133] Shi G, Shi X, O'Connell M, Yu R, Azizzadenesheli K, et al. Neural lander: Stable drone landing control using learned dynamics. In Proceedings of 2019 International Conference on Robotics and Automation (ICRA), Montreal, Canada, 20–24 May 2019, pp. 9784–9790.